Author’s response to reviews

Title: Single center experience with laparoscopic adrenalectomy on a large clinical series.

Authors:

Giovanni Conzo (giovanni.conzo@unicampania.it)
Claudio Gambardella (claudiog86@hotmail.it)
Giancarlo Candela (giancarlo.candela@unicampania.it)
Alessandro Sanguinetti (sanguinettiiale@gmail.com)
Andrea Polistena (apolis74@yahoo.it)
Guglielmo Clarizia (clarizia.guglielmo@libero.it)
Renato Patrone (dott.patrone@gmail.com)
Francesco Di Capua (francesc.dicapua@gmail.com)
Chiara Offi (chiara.o@live.com)
Sergio Iorio (sergio.iorio@unicampania.it)
Giuseppe Bellastella (giuseppe.bellastella@unicampania.it)
Daniela Pasquali (daniela.pasquali@unicampania.it)
Annamaria De Bellis (annamaria.debellis@unicampania.it)
Antonio Sinisi (antonio.sinisi@unicampania.it)
Nicola Avenia (nicola.avenia@unicampania.it)

Version: 1 Date: 04 Jun 2017

Author’s response to reviews:

Dear Editor, thank you for the advices you gave me about my manuscript. I followed your instructions and corrected what was missing.

The ethics and consent to publish are present in the manuscript.

I added the notes in the tables.
The thirty procedures were considered basing on literature.